%% if you are submitting an initial manuscript then you should have submission as an option here
%% if you are submitting a revised manuscript then you should have revision as an option here
%% otherwise options taken by the article class will be accepted
\documentclass[finalversion]{FPSAC2018}
\articlenumber{63}
%% but DO NOT pass any options (or change anything else anywhere) which alters page size / layout / font size etc

%% note that the class file already loads {amsmath, amsthm, amssymb}

%\usepackage{showkeys}

\newtheorem{thm}{Theorem}[section]
\newtheorem{lem}[thm]{Lemma}
\newtheorem{definition}[thm]{Definition}
\newtheorem{corollary}[thm]{Corollary}
\newtheorem{conjecture}[thm]{Conjecture}
\newtheorem{prop}[thm]{Proposition}


\DeclareMathOperator{\sgn}{sgn}

\usepackage{lipsum}

%% define your title in the usual way
\title{Action of the symmetric group on the free LAnKe: a CataLAnKe Theorem}

%% define your authors in the usual way
%% use \addressmark{1}, \addressmark{2} etc for the institutions, and use \thanks{} for contact details

%\author[Optional shorter names]{Longer names me\thanks{\href{mailto:hello@world.c}{hello@world.c}. Longer names me was partially supported by Grant 2017.11.14.$\partial$\;supp.}\addressmark{1}, \and you\addressmark{2}}

\author[Tamar Friedmann, Philip Hanlon, Richard P. Stanley, and Michelle
L. Wachs]{Tamar Friedmann\thanks{\href{mailto:tamarf1@yahoo.com}{tamarf1@yahoo.com}}\addressmark{1},
Philip Hanlon\addressmark{2},
Richard P.
Stanley\thanks{\href{mailto:rstan@math.mit.edu}{rstan@math.mit.edu}.
Supported in part by NSF grant DMS 1068625.}\addressmark{3,4}, \\
and Michelle L. Wachs\thanks{\href{mailto:wachs@math.miami.edu}{wachs@math.miami.edu}. Supported in part by NSF grants DMS 1202755, DMS 1502606, and by Simons Foundation grant \#267236.}\addressmark{4}}
%% then use \addressmark to match authors to institutions here
\address{\addressmark{1}Department of Mathematics and Statistics, Smith College \\ \addressmark{2}Department of Mathematics, Dartmouth College \\
\addressmark{3}Department of Mathematics, MIT \\
\addressmark{4}Department of Mathematics, University of Miami}

%% put the date of submission here
\received{November 14, 2017}

%% leave this blank until submitting a revised version
\revised{April 11, 2018}

%% put your English abstract here, or comment this out if you don't have one yet
%% please don't use custom commands in your abstract / resume, as these will be displayed online
%% likewise for citations -- please don't use \cite, and instead write out your citation as something like (author year)
%\abstract{\lipsum[1]}

\abstract{We initiate a study of the representation of the symmetric group on the multilinear component of an $n$-ary generalization  of the free Lie algebra, which we call a free LAnKe.  Our central result is that  the representation of the symmetric group $S_{2n-1}$ on the multilinear
  component of the free LAnKe with $2n-1$ generators is given by an irreducible representation
whose dimension is the $n$th Catalan number. This leads to a more general result on eigenspaces of 
  a certain linear operator.  A decomposition,  into  irreducibles,  of  the representation of $S_{3n-2}$ on the multilinear component the free LAnKe with $3n-2$ generators   is  also presented.  We also obtain a new presentation of  Specht modules of shape $\lambda$, where $\lambda$ has strictly decreasing column lengths, as a consequence of our eigenspace result.  
  }




%% put your keywords here, or comment this out if you don't have them yet
\keywords{Free Lie algebra, Specht modules, Catalan numbers}

%% you can include your bibliography however you want, but using an external .bib file is STRONGLY RECOMMENDED and will make the editor's life much easier
%% regardless of how you do it, please use numerical citations, ie. [xx, yy] in the text

%% this sample uses biblatex, which (among other things) takes care of URLs in a more flexible way than bibtex
%% but you can use bibtex if you want
\usepackage[backend=bibtex]{biblatex}
\addbibresource{63-Friedmann-Hanlon-Stanley-Wachs.bib}
%% note the \printbibliography command at the end of the file which goes with these biblatex commands


\usepackage{tikz-cd}
\usetikzlibrary{cd}
\usetikzlibrary{decorations.pathmorphing}

\usepackage[vcentermath,enableskew]{youngtab}
\usepackage{rotating}
%\usepackage{slashbox}
\usepackage{diagbox}
\usepackage{array}
\newcolumntype{L}{>{$}l<{$}} % math-mode version of "l" column type
\newcolumntype{C}{>{$}c<{$}} % math-mode version of "c" column type
\usepackage{ytableau}


\newcommand{\no}{\noindent}

\newcommand{\be}{\begin{equation}} 
\newcommand{\bea}{\begin{eqnarray}}
\newcommand{\ee}{\end{equation}}
\newcommand{\beas}{\begin{eqnarray*}}
\newcommand{\eea}{\end{eqnarray}}
\newcommand{\eeas}{\end{eqnarray*}}
\newcommand{\non}{\nonumber}
\newcommand{\lie}{\mathrm{Lie}}

\def\G{\Gamma}
\def\S{{\bf S}}
\def\R{{\bf R}}
\def\Z{{\mathbb Z}}
\def\C{{\mathbb C}}
\def\P{{\bf P}}
\def\G{\Gamma}
\def\N{{\cal N}}
\def\L{{\cal L}}
\newcommand{\sm}{SU(3)\times SU(2)\times U(1)}
\newcommand{\ps}{SU(2)_L\times SU(2)_R\times SU(4)_C}
\newcommand{\su}{\mathfrak{su}(2)}
\newcommand{\cz}{\C^3/\Z_3}
\newcommand{\CZ}{\C^n/\Z_n}
\newcommand{\g}{\mathfrak g}
\newcommand{\h}{\mathfrak h}


\begin{document}
\begin{thebibliography}{99999999}


\bibitem[BL]{BL}
  J.~Bagger and N.~Lambert,
  {\it Three-Algebras and N=6 Chern-Simons Gauge Theories},
  Phys.\ Rev.\  D {\bf 79} (2009), 025002, 8 pp.,
  [arXiv:0807.0163 [hep-th]]. 
 
 
 \bibitem[Ba]{Ba}
H.~Barcelo,
{\it On the action of the symmetric group on the free {L}ie algebra and
  the partition lattice},
J. Combin. Theory Ser. A, {\bf 55}(1) (1990), 93--129.
 
 \bibitem[BF]{BF}
S. ~Brauner and T. ~Friedmann, {\it One {G}arnir to rule them all: on {S}pecht modules and the {C}ata{LA}n{K}e theorem}, [arXiv:1803.1091 [math.CO]].


\bibitem[Co]{Co} F.~Cohen, 
{\it On configuration spaces, their homology, and Lie algebras}, J. Pure and Applied Algebra {\bf 100} (1995) 19--42. 

\bibitem[DT]{DT} Y.L.~Daletskii and L.A.~Takhtajan,
{\it Leibniz and Lie algebra structures for Nambu algebra},  
Lett. Math. Phys. {\bf 39} (1997),  127--141.

\bibitem[DI]{DI} J. De Azcarraga and J.  Izquierdo, {\it  $n$-ary algebras: A review with applications},  J.Phys. A {\bf 43} 293001-1-117 (2010).


  \bibitem[Fi]{Fi}
  V.T.~Filippov, 
  {\it n-Lie algebras}, 
  Sibirsk.\ Math.\ Zh. {\bf 26} (1985), 126--140. 


\bibitem[Fr]{Fr} T.~Friedmann, {\it Orbifold Singularities, Lie Algebras of the Third Kind (LATKes), and Pure Yang-Mills with Matter}, J. Math. Phys. {\bf 52} (2011), 022304,  [arXiv:0806.0024 [math.AG]].

\bibitem[FHSW1]{FHSW1} T. Friedmann, P. Hanlon, R. Stanley, and M. Wachs, {\it On a generalization of Lie(k): a CataLAnKe theorem}, [arXiv:1710.00376 [math.CO]].

\bibitem[FHSW2]{FHSW2} T. Friedmann, P. Hanlon, R. Stanley, and M. Wachs, {\em in progress}.


\bibitem[Fu]{Fu} W. Fulton, {\it Young Tableaux}, London Mathematical Society, Student Texts {\bf 35}, Cambridge University Press, Cambridge, 1996.

 
 \bibitem[Gu]{Gu} A.~Gustavsson,
  {\it Algebraic structures on parallel M2-branes},
  Nucl.\ Phys.\  B {\bf 811} (2009), 66--76,
  [arXiv:0709.1260 [hep-th]].

\bibitem[Ha]{Han} P. Hanlon, {\it The fixed-point partition lattices}, Pacific J. Math. {\bf 96} (1981), 319--341. 

\bibitem[HW]{HW} P. Hanlon and M.L. Wachs, {\it On Lie $k$-algebras}, Adv.  Math. {\bf 113}  (1995), 206--236. 

 \bibitem[Ka]{Ka}   S.~M.~Kasymov,
  {\it On a theory of n-Lie algebras},
Algebra\ Logika\ {\bf 26} (1987),  277--297.
 % Algebra\ i\ Logika\ {\bf 26} (1987),  277--297.
  


%\bibitem[Kr]{Kr} W. Kraskiewicz, {\it Reduced decompositions in Weyl groups}, Europ. J. Combinatorics {\bf 16} (1995), 293-313.

\bibitem[KW]{KW} W. Kraskiewicz and J. Weyman,  {\it Algebra of coinvariants and the action of a Coxeter element}, Bayreuth. Math. Schr. {\bf 63} (2001), 265--284.


\bibitem[KS]{KS} B. Kol and R. Shir, {\it Color Structures and Permutations}, J. High Energ. Phys. (2014) 2014: 20.

  \bibitem[Li]{Li}
  W.~X.~ Ling, 
  {\it On the structure of n-Lie algebras},
  PhD Thesis, Siegen, 1993.



\bibitem[Re]{Re} C. Reutenauer,  {\it Free Lie Algebras}, London Mathematical Society Monographs, New Series {\bf 7},  Clarendon Press, Oxford University Press, New York, 1993.





\bibitem[St]{St} R.P. Stanley, {\it Some aspects of groups acting on finite posets},  J. Combin.Theory, Ser. A {\bf 32} (1982), 132--161.



  \bibitem[Ta]{Ta}
L.A.~Takhtajan,
{\it On foundation of the generalized Nambu mechanics},
Comm. Math. Phys. {\bf 160} (1994), 295--315.


\bibitem[Wa]{Wa}
M.L. Wachs,
{\it On the (co)homology of the partition lattice and the free Lie
  algebra},
Discrete Math. {\bf 93} (1998), 287--319.





\end{thebibliography}


\maketitle
%% note that you DO NOT have to put your abstract here -- it is generated by \maketitle and the \abstract and \resume commands above


\section{Introduction}


Lie algebras are defined as vector spaces equipped with an
antisymmetric commutator and a Jacobi identity. They are a cornerstone
of mathematics and have applications in a wide variety of areas of
mathematics as well as physics. Also of fundamental importance is the
free Lie algebra, a natural mathematical construction central in the
field of algebraic combinatorics. The free Lie algebra has beautiful
dimension formulas;  elegant bases in terms of binary trees;
and connections to the shuffle algebra, Lyndon words, necklaces, Witt
vectors, the descent algebra of $S_n$, quasisymmetric functions,
noncommutative symmetric functions, and the lattice of set
partitions. 

It was the search for a tool to solve a problem in string theory that led
the first author to define a generalization of Lie algebras, called a
``LAnKe'',
or Lie Algebra of the $n^{th}$ Kind 
 \cite{Fr}\footnote{The $n=3$ case of LAnKe is called a LATKe in \cite{Fr}}. 
 Specifically, the LAnKe arose as the algebraic object that plays a central role in generalizing a relation between codimension-4 singularities and ADE Lie algebras to a relation between higher-codimension singularities and algebraic objects. The singularities appear in the extra dimensions of string/M theory and affect the interactions seen in the 4-dimensional physical world. 


In this extended abstract of \cite{FHSW1}, we focus on generalizing the free Lie algebra to the free LAnKe and studying the representation  of the symmetric group on its multilinear component.   Our study generalizes the well-known representation $\lie(k)$.  The free LAnKe is based on an $n$-ary generalization of the Lie bracket.  We define $\rho_{n,k}$ to be the representation of the symmetric group $S_m$ on the multilinear component of the free LAnKe on $m$ generators, where $m:=n(k-1) - k+2$. 
Our central result is that  $\rho_{n,3}$ is isomorphic to  the Specht module
$S^{2^{n-1}1}$, whose dimension is the $n^{th}$ Catalan number.  We also present the result, whose proof will appear in \cite{FHSW2}, that $\rho_{n,4}$ is isomorphic to a sum of two Specht modules.  

An explicit $S_m$-isomorphism from $\rho_{n,3}$ to $S^{2^{n-1}1}$ can be obtained from  presentations of  free LAnKes and   Specht modules following from results in \cite{DI} and  \cite{Fu}, respectively.   In this paper  the relationship between $\rho_{n,3}$ and $S^{2^{n-1}1}$ is placed in a  more general setting.  The $S_m$-module $\rho_{n,k}$ has a  presentation of the form $V_{n,k}/ R_{n,k}$, where $V_{n,k}$ is generated by $n$-bracketed permutations involving an   $n$-bracket  that is  antisymmetric only, and $R_{n,k}$ is the submodule of $V_{n,k}$ generated by the generalized Jacobi relations used in the definition of  LAnKe.  We consider a natural linear operator on $V_{n,3}$ whose $0$th eigenspace  is isomorphic to $\rho_{n,k}$.  We show that  all the  eigenspaces  are  irreducible of the form $S^{2^{i}1^{2n-1-2i}}$.   Techniques from this proof also play a role in the proof  of the above mentioned decomposition for $\rho_{n,4}$ obtained in \cite{FHSW2}.

 


 In Section~\ref{Defn} we introduce 
LAnKes 
and their relation to orbifold singularities (note that the rest of this extended abstract can be read independently of this section except for Definition \ref{lankedef}). In Section~\ref{franke} we introduce the free LAnKe and define the $S_m$-module $\rho_{n,k}$ which  generalizes $\lie (k)$. Our results for $\rho_{n,k}$ where $k=3,4$, are also included in Section~\ref{franke}.  In Section~\ref{catalanrep} we present our  
result on eigenspaces of the above mentioned linear operator on $V_{n,3}$, which yields the $k=3$ result.
In Section~\ref{stair} we discuss   the presentations of LAnKes and Specht modules that yield the explicit isomorphism for $k=3$.  We also discuss a  new presentation of Specht modules of shape $\lambda$, where $\lambda$ has strictly decreasing column lengths. This is a consequence of a generalization of our eigenspace result.

\section{The definition of the
 LAnKe}\label{Defn}

How was the definition of the LAnKe obtained, and what  is the LAnKe? This section  provides a roadmap that begins with the relation between singularities of codimension 4  and Lie algebras, and culminates in the definition of the LAnKe. 

\subsection{Orbifold singularities and Lie algebras}


 Consider an orbifold singularity of codimension 4, of the form
 $\C^2/\Gamma$ where $\Gamma$ is a finite subgroup of $SU(2)$  acting
 naturally on the complex plane $\C^2$.  
For example, $\Gamma = \Z_n$ is generated by the diagonal $2\times 2$
complex matrix diag$\{e^{2\pi i/n},e^{-2\pi i/n}\}$,  
and its action on $\C^2$ is given by $(x,y)\rightarrow (xe^{2\pi i/n},
ye^{-2\pi i/n})$. The action fixes the origin $(x,y)=(0, 0)$ while
acting freely on the rest of the plane. The resolution of the
singularity at the origin is a collection of intersecting 2-spheres
known as the exceptional divisor of the singularity. When viewed as a
graph, the intersecting 2-spheres form the Dynkin diagram of the Lie
algebra $\mathfrak{sl}_n(\C)$, known as a Lie algebra of type $A$;
further,  the intersection matrix $I_{ij}$ of the exceptional divisor
is (the minus of) the Cartan matrix $C_{ij}$ of the corresponding Lie
algebra. 
Similarly, the other finite discrete subgroups of $SU(2)$ lead to
singularities at the origin whose resolutions correspond to the Dynkin
diagrams of Lie algebras of types $D$ and $E$, with the same relation
between the intersection numbers and the Cartan matrix.  Due to their
ADE classification, the codimension-4 orbifold singularities are known
as ADE singularities.   


The physical picture that involves codimension-4 singularities is as follows: the ADE singularities appear in the extra dimensions, and lead to a principal $G_{ADE}$-bundle over space-time crossed with the locus of the singularity. The group $G_{ADE}$ is the Lie group of the corresponding ADE Lie algebra and it plays a central role in {\em gauge theories}, which are essentially a family of symmetries together with a physical model that can describe the interactions in the four-dimensional world. 





What happens when the singularities have higher codimension? 
Can the relation between singularities and Lie algebras be generalized to codimension-$2n$ singularities? 
That is, given an orbifold singularity  $\C^n/\Gamma$, where $\Gamma$ is a finite, discrete subgroup of $SU(n)$, is there an algebraic object that can be associated with the singularity in a way analogous to that in which ADE Lie algebras are associated with ADE singularities?
This question is illustrated in 
the following diagram.

\be \label{diagram}
\begin{tikzcd}
\C ^2/\Gamma_{ADE} \arrow[r, leftrightarrow] \arrow[d, dashed] &    \mbox{ADE Lie algebra}  \arrow[d, dashed] \\
\C^n/\Gamma   \arrow[r, leftrightsquigarrow]&  ?\\
\end{tikzcd}
\ee
We discuss the answer in the next subsection. Here, we mention that just like the $n=2$ case, the $n=3$ case -- with singularities of codimension 6 -- has applications in string/M theory; for further details, see \cite{Fr}.


\subsection{The Commutator-Intersection-Relations and the LAnKe}

To answer the question of the previous subsection, we look back at the
traditional Lie algebra case. Recall that for any Lie algebra, the
entries of the Cartan matrix, $C_{ij}$, play a central role in the
Chevalley--Serre relations that contain the commutation relations of the
generators of the Lie algebra.  In our setting, said entries  are equal
to elements of  the intersection matrix of the associated singularity
(with a minus sign). Replacing  $C_{ij}$ by $-I_{ij}$ in the  Chevalley--Serre relations leads to a fundamental reinterpretation. The relations become:
\begin{align}
&[H_i, H_j]=0 &  [X_i, Y_j] = \delta _{ij} H_j \; ; \hskip .3cm \non
\\ \label{CIRs}
&[H_i, X_j]=-I_{ij}X_j &  [H_i, Y_j]=I_{ij}Y_j \; ; \hskip .3cm 
\\
&\mbox{ad}(X_i)^{1+I_{ij}}(X_j)=0 & \mbox{ad}(Y_i)^{1+I_{ij}}(Y_j)=0\; . \hskip .3cm\non
\end{align}
Looking at these relations afresh reveals that actually, they  provide all commutators of the Lie algebra {\em in terms of} the intersection numbers of the exceptional divisor of the corresponding singularity! 
These {\em Commutator-Intersection Relations} (CIRs) will now be generalized to the codimension 6 case, and then to codimension $2n$. 


We begin by generalizing the intersection matrix. For codimension-6 orbifolds, the intersection form of the exceptional divisor is no longer a matrix of size $r\times r$, where $r$ is both the rank of the algebra and the number of irreducible components in the exceptional divisor. Rather, it is  a hyper-matrix of dimension $r\times r\times r$. This follows from the fact that in order to obtain a zero-dimensional space from which we can define an intersection number, it is required that we take triple intersections. For general $n$, we need $n$-fold intersections. 

Now consider equations (\ref{CIRs}), in particular $[H_i, X_j]=-I_{ij}X_j$. The number of indices in the intersection matrix (two: $i$ and $j$) is the same as the number of entries in the commutator (two: $H_i$ and $X_j$). For codimension 6, the number of indices in the intersection form is 3, and for codimension $2n$ this number is $n$. This leads to the concept of a commutator of 3, or $n$, entries. All we need now is antisymmetry and a generalized Jacobi identity, and we will have obtained a mathematical object generalizing Lie algebras and replacing the question mark in diagram (\ref{diagram}).

\begin{definition}\label{lankedef}  A {\textbf {\textit  {Lie algebra
$\cal L$ of the $n$-th kind (``LAnKe''
 )}}}   is a vector space equipped with an $n$-linear bracket
\[ [\cdot, \cdot , \hskip .7cm , \cdot] : \times ^n \L \rightarrow \L ~\]
that satisfies the following antisymmetry relation for all $\sigma$ in the symmetric group $S_n$:
\begin{equation} \label{antsym} [x_1,\dots,x_n] = \sgn(\sigma) [x_{\sigma(1)},\dots, x_{\sigma(n)}] \end{equation}
and the following generalization of the Jacobi identity:
\begin{align}
\label{type1} &[[x_1, x_2, \ldots,  x_n], x_{n+1},\ldots , x_{2n-1}]
\\ \non&=\sum_{i=1}^{n} [x_1, x_{2}, \ldots , x_{i-1},[ x_{i}, x_{n+1}, \ldots , x_{2n-1}], x_{i+1}, \ldots , x_{n}],
\end{align}
for all $x_1,x_2,\dots,x_{2n-1} \in \mathcal L$.

\end{definition}

The definition of the LAnKe with the above-described motivation was
given in \cite{Fr}. The same algebraic structure arose before in other
contexts as well; see \cite{Fi, Ta, DT, Ka, Li, BL, Gu}. 
A different generalization of Lie algebras that also involves $n$-ary brackets appeared in the 1990's in work of Hanlon and Wachs \cite{HW}. 

We now proceed to use this definition in our generalization of the free Lie algebra. 

\section{The free LAnKe}\label{franke}

There are beautiful results on the free Lie algebra involving  the representation of the symmetric group $S_k$ on the multilinear component of the free Lie algebra with $k$ generators. This representation is known as $\lie(k)$. The LAnKe provides an opportunity to generalize both the free Lie algebra itself and the representation $\lie(k)$. To set the stage, we review the free Lie algebra and $\lie(k)$ first (see \cite{Re}).  


\subsection{The free Lie algebra and $\lie(k)$}\label{fla}



 Let $X:=\{x_1, x_2, \ldots , x_k\}$ be a set of generators. Then the
 multilinear component of the free Lie algebra on $X$ is the subspace
 spanned by bracketed ``words'' where each generator in $X$ appears exactly once.  For example,
 $[[x_1, x_3],[[x_4, x_5],x_2]]$ is such a bracketed word when $k=5$,
 while $[[x_1, x_3],[[x_1, x_5],x_3]]$ is not. A certain type of
 bracketed word in the multilinear component has the form 
\be  \label{FLAbasis} [\cdots [[[x_{\sigma(1)}, x_{\sigma(2)}],x_{\sigma(3)}], x_{\sigma(4)}],\dots , x_{\sigma(k)}], \hskip 1cm \sigma\in S_k .\ee
Bracketed words that are not of this type, such as $[[x_1, x_3],[[x_4,
      x_5],x_2]]$, can be shown to be linear combinations of the
bracketed words of the form in   (\ref{FLAbasis}) using
iterations of the Jacobi identity. Furthermore, if we restrict  to the
permutations that satisfy $\sigma(1)=1$, then the words of the form in
(\ref{FLAbasis}) form a basis for the vector space. This vector space
admits a natural $(k-1)!$-dimensional representation of $S_k$ denoted
$\lie(k)$.  

The decomposition of $\lie(k)$ into irreducibles is given by the following result.

\begin{thm}[Kraskiewicz and Weyman \cite{KW}] \label{liek}
 Let $i\ge 1$ and $k\ge 2$ be relatively prime.  For each $\lambda \vdash k$, the multiplicity of the Specht module $S^\lambda$ in $\lie(k)$ is equal to the number of standard Young tableaux of shape $\lambda$ and of major index congruent to $i \mod k$. 
\end{thm}



Interestingly, $\lie(k)$ appears in a variety of other contexts, such
as the top homology of the lattice of set partitions in work of
Stanley \cite{St}, Hanlon \cite{Han}, Barcelo \cite{Ba}, and Wachs
\cite{Wa}, in the  homology of configuration spaces of $k$-tuples of
distinct points in Euclidean space in work of Cohen \cite{Co}, and in scattering amplitudes in gauge theories in work of Kol and Shir \cite{KS}.  




\subsection{The free LAnKe  and $\rho_{n,k}$}

The following generalizes the standard definition of a free Lie algebra.

\begin{definition} Given a set $X$, the  {\bf free LAnKe on $X$ } is a LAnKe $\L$ together with a mapping $i:X\rightarrow \L$ with the following universal property: for each LAnKe $\cal K$ and each mapping $f:X\rightarrow \cal K$, there is a unique LAnKe homomorphism $F:\L\rightarrow \cal K$ such that $f=F\circ i$.
\end{definition}


Similar to the free Lie algebra, a LAnKe is free on X if it is generated
by all possible $n$-bracketings of elements of $X$, and if the only
possible relations existing among these bracketings are consequences
of $n$-linearity of the bracketing, the antisymmetry of the bracketing (\ref{antsym}), and the generalized Jacobi identity (\ref{type1}).


The multilinear component of the free LAnKe on $m$ generators is  spanned by bracketed permutations; that is, bracketed words where each generator appears exactly once. 
We consider two variables: $n$, the number of entries in a given
bracket (so $n=2$ for Lie algebras), and $k$, the number of 
brackets plus 1.  The number of generators for the  multilinear
component is then $kn-n-k+2$, an expression symmetric in $n$ and
$k$. For example, an element of the form $[\cdot \cdot [\cdots  ]]$
has $n=3$, $k=3$, and $3\cdot 3-3-3+2=5$ generators; $[\cdot
  \cdot[\cdot \cdot[\cdots]]]$ has $n=3$, $k=4$, and $4\cdot
3-4-3+2=7$ generators.  

The object we study here is the  representation of $S_{m}$ on the multilinear component of the free LAnKe on $m:={kn-n-k+2}$ generators. We  denote this representation by $\rho_{n,k}$, and  view $( \rho_{n,k} )$ as an array of representations, with $n,k \geq 2$. 

\begin{figure}[!t]

\begin{minipage}{\linewidth}
\small



\begin{tabular}{|C|| C|C|C|C|C||}


\hline 
&\multicolumn{5}{c|}{\mbox{{\bf Table 1: What is known  about the representations $\rho_{n,k}$}}}\\ 
\hline 


${\backslashbox{$n$ \kern-1em}{ \kern-1em $k$}}$
 &2&3&4&5&k \\ \hline 
&S_2&S_3&S_4&S_5&S_{k}  \\
&[..]&[.[..]]&[.[.[..]]]&[.[.[.[..]]]]
&\\
2&\yng(1,1)&\yng(2,1)& \yng(2,1,1) \oplus  \yng(3,1) &
\begin{array}{c} 32\oplus  41\\  \oplus  21^3  \oplus  31^2 \oplus  2^21 \end{array} &Lie(k)\\ 
&1&2&
6&24&(k-1)!\\ \hline
&S_3&S_5&S_7&S_9&S_{2k-1}\\
&[...]&[..[...]]&[..[..[...]]]&[..[..[..[...]]]]&\\
3&\yng(1,1,1)&\yng(2,2,1)&\yng(3,2,1,1)\oplus \yng(3,3,1)&\begin{array}{c}432\oplus  4^21 \\ \oplus  421^3 \oplus  431^2 \oplus  3^21^3\\ \oplus  42^21\oplus  32^3\end{array} & \rho_{3,k}\\
&1&5&
56&1077 &\\ \hline
&S_4&S_7&S_{10}&S_{13}&S_{3k-2}\\
&[....]&[...[....]]&[...[...[....]]]&[...[...[...[....]]]]&\\
4&\yng(1,1,1,1)&\yng(2,2,2,1)&\yng(3,3,2,1,1)\oplus  \yng(3,3,3,1)& 
\rho_{4,5} & \rho_{4,k}\\
&1&14&
660&& \\ \hline
&S_n&S_{2n-1}&S_{3n-2}&S_{4n-3}&S_{nk-n-k+ 2}\\
n&1^n&2^{n-1}1&3^{n-2}21^2\oplus  3^{n-1}1 &
\rho_{n,5}
& \rho_{n,k}\\ 
&1&{1\over n+ 1}{2n \choose n}&
{4\over \prod_{i=1}^3 (n+ i)} {3n \choose n,n,n}&&\\
&&&&&\\
\hline
\end{tabular}



 
\end{minipage}

\end{figure}

Table 1 summarizes what we know  about the decomposition of $\rho_{n,k}$ into irreducibles. The Young diagrams in the table stand for Specht modules of the indicated shape.   
The first row $\rho_{2,k}$,  $k\geq 2$, is the Lie representation $\lie(k)$.  The sign representations that appear in the first column, $k=2$,  trivially follow from the antisymmetry of the bracket.  The second column $\rho_{n,3}$, $n\geq 2$, follows from Theorem~\ref{catalanke} below and the third column $\rho_{n,4}$, $n\geq 2$, follows from Theorem~\ref{k4} below.  The first three columns suggest that $\rho_{n,k}$ can be obtained from $\rho_{n-1,k}$ by adding a row of length $k-1$ to each irreducible of $\rho_{n-1,k}$.  However the entry $\rho_{3,5}$, whose expansion follows from general results of the authors in \cite{FHSW2}, shows that this does not hold when $k =5$.  Relationships between $\rho_{n,k}$ and $\rho_{n-1,k}$ are explored in \cite{FHSW2}.


\begin{thm} \label{catalanke}   For all $n \ge 2$, the representation $\rho _{n,3}$ is given by the Specht module  $S^{2^{n-1}1}$, whose dimension is  the $n^{th}$ Catalan number ${1\over n+1}{2n\choose n}$.   
\end{thm}

An explicit $S_{2n-1}$-module isomorphism between  the CataLAnKe
representation $\rho _{n,3}$ and the Specht module $S^{2^{n-1}1}$ can be
obtained from an  alternative generalized Jacobi relation in \cite{DI}
and a presentation   for any Specht module $S^\lambda$ in \cite{Fu}; see Section~\ref{stair}.   In the next section we show that Theorem~\ref{catalanke} can be viewed as a special case of a result on eigenspaces of a certain operator.

The following result is obtained by the authors in \cite{FHSW2}.  Techniques discussed in the next section play a role in the proof.
\begin{thm} \label{k4} For all $n \ge 2$, the following $S_{3n-2}$-module isomorphism holds,
$$\rho_{n,4} \cong S^{3^{n-2}21^2} \oplus S^{3^{n-1}1} .$$
\end{thm}




\section{The CataLAnKe representation and a linear operator}\label{catalanrep}

In this section we consider an operator whose null space is isomorphic to the CataLAnKe representation $\rho_{n,3}$. We show that all the eigenspaces of this operator are irreducible and use this to recover Theorem~\ref{catalanke}.

 Let $V_{n,3}$ be the multilinear component of the  vector space generated by all possible  $n$-bracketed words on $[2n-1]$, subject only to  antisymmetry of the brackets  given in (\ref{antsym}) (but not to the generalized Jacobi identity (\ref{type1})).  That is, $V_{n,3}$ is the subspace generated by 
$$u_\tau:= [ [\tau_1, \ldots,  \tau_{n}],  \tau_{n+1}, \ldots ,\tau_{2n-1}],$$
where $\tau \in S_{2n-1}$, $\tau_i=\tau(i)$ for each $i$, and $[ \cdot, \dots, \cdot ]$ is the antisymmetric $n$-linear bracket (that does not satisfy the generalized Jacobi relation).

The symmetric group $S_{2n-1}$ acts on generators of $V_{n,3}$ by the following action: for $\sigma,\tau  \in S_{2n-1}$, 
$$\sigma u_\tau = u_{\sigma\tau}.$$
This induces a representation of $S_{2n-1}$ on $V_{n,3}$ since the action respects the antisymmetry relation.

For each $n$-element subset $S:= \{a_1,\dots,a_n\}$ of $[2n-1]$, let 
$$v_S =  [ [a_1, \ldots , a_{n}], b_1, \ldots ,b_{n-1} ],$$
 where  $ \{b_1 , \cdots , b_{n-1} \}= [2n-1] \setminus S $, and the $a_i$'s and $b_i$'s are in increasing order.   Clearly, 
\begin{equation} \label{basiseq} \left \{v_S : S \in \binom{[2n-1]}{n}\right \}\end{equation} is a basis for $V_{n,3}$.  Thus $V_{n,3}$ has dimension $\binom{2n-1}{n}$.

 For each $S \in \binom{[2n-1]}{n}$, use the generalized Jacobi identity (\ref{type1}) to define   the  relation
\be \label{jacobirel} R_S=  v_S \, -\sum_{i=1}^n [a_1, \ldots ,a_{i-1}, [a_i, b_1, \ldots , b_{n-1} ], a_{i+1}, \ldots , a_{n}], \hskip 3cm\ee
where $a_1 <\dots < a_n $ and $b_1 < \dots <b_{n-1}$ are as in the previous paragraph.
Let $R_{n,3}$ be the subspace of $V_{n,3}$ generated by the $R_S$.    Then as $S_{2n-1}$-modules   

\begin{equation} \label{Req} V_{n,3}/R_{n,3} \cong \rho_{n,3}.\end{equation}

We consider the linear operator $\varphi: V_{n,3} \to V_{n,3}$ defined on  basis elements by
$$\varphi(v_S) = R_S.$$  It is not difficult to see that this is an $S_{2n-1}$-module isomorphism.
We show that as $S_{2n-1}$-modules, 
\begin{equation} \label{nulleq} \mbox{ null } \varphi \cong \rho_{n,3}.\end{equation} Hence   Theorem~\ref{catalanke} says that the null space is isomorphic to the Specht module $S^{2^{n-1}1}$. The next result generalizes this to all the eigenspaces of $\varphi$. 

 \begin{thm} \label{eigenth} There are $n$ distinct eigenvalues of  $\varphi$, which are given by 
$$w_i:= 1+(n-i)(-1)^{n-i},$$ for $i=0,1,\dots,n-1$.  Moreover, if $E_i$ is the eigenspace corresponding to $w_i$ then as $S_{2n-1}$-modules
$$E_i \cong S^{2^i1^{(2n-1)-2i}},$$ for each $i =0,1,\dots, n-1$.
 \end{thm}   



Note that we can recover Theorem~\ref{catalanke} as a consequence of  Theorem~\ref{eigenth} and  (\ref{nulleq}) since $w_i = 0$ when $i=n-1$.

\begin{proof}[Proof idea]\footnote{See \cite{FHSW1} for a full-length manuscript with the complete proof of Theorem \ref{eigenth}.} 
We use  Young's rule to show
\[V_{n,3} \cong \bigoplus _{i=0}^{n-1} S^{2^i1^{2n-1-2i}}.\]
Since there are no multiplicities, it follows from Schur's lemma that $\varphi$ acts as a scalar on each irreducible submodule. 
It remains to determine said scalar for each irreducible $S^{2^i1^{2n-1-2i}}$. 

 To compute the scalar, we start by letting $t$ be the Young tableau of shape $2^i1^{2n-1-2i}$ whose first column is
 $1,2,\dots,n, n+i+1,n+i+2, \dotsm 2n-1$ and whose second column is $n+1,n+2,\dots,n+i$.
 \no Let $C_t$ be the column stabilizer of $t$ and let $R_t$ be the row stabilizer.
Recall that the Young symmetrizer associated with $t$ is defined by $$e_t:=\sum_{\alpha \in R_{t}}  \alpha  \\  \sum_{\beta \in C_t} \sgn(\beta) \beta $$ 
and that the Specht module $S^{2^i1^{2n-1-2i} }$ is the submodule of the regular representation  $\C S_{2n-1}$ spanned by $\{\tau e_t : \tau \in S_{2n-1} \} $.  

 Now set $T:=[n]$.  We show that $e_tv_T \ne 0$. 
Let $\psi:\C S_{2n-1} \to V_{n,3}$ be the $S_{2n-1}$-module
homomorphism defined  by $\psi(\sigma) = \sigma v_T$. 
Now consider the  restriction of $\psi$
to the Specht module  $S^{2^i1^{2n-1-2i} }$.    By the irreducibility
of  the Specht module and  the fact that $e_t v_T \ne 0$, this
restriction  is  an  isomorphism from  $S^{2^i1^{2n-1-2i} }$ to the
subspace of $V_{n,3}$ spanned by $\{\tau e_t v_T : \tau \in S_{2n-1}
\} $.  This subspace is therefore the unique subspace of $V_{n,3}$
isomorphic to $S^{2^i1^{(2n-1)-2i}}$.    Thus
$\varphi(e_tv_T) = c e_tv_T$ for some scalar $c$.  By computing the coefficient of $v_T$ in 
$\varphi(e_tv_T)$ and in $e_tv_T $ we show that $c= w_i$.
\end{proof}

\section{Alternative presentations}\label{stair}



For each partition $\lambda= (\lambda_1 \geq\dots \geq \lambda_l)$ of $m$,
let $\mathcal T_\lambda$ be the set of Young tableaux of shape
$\lambda$.
Let $M^\lambda$ be the vector space generated by  
$\mathcal T_\lambda$ subject only to column relations, which are of
the form $t+s$, where $s$ is obtained from $t$ by switching two
entries in the same column.  
Given $t \in \mathcal T_\lambda$, let $\bar t$ denote the coset of $t$
in $M^\lambda$. These cosets, which are called {\it column tabloids}, generate $M^{\lambda}$.
 The symmetric group $S_m$ acts on $\mathcal T_\lambda$ by replacing
 each entry of a tableaux by its image under the permutation in $S_m$.
 This induces a representation of $S_m$ on $M^\lambda$. 

  
There are various different presentations of $S^\lambda$ in the literature, which involve the column relations and Garnir relations.  Here we are interested in a presentation of $S^\lambda$ discussed in Fulton \cite[Section 7.4]{Fu}.
The Garnir relations are of the form
$\bar t-\sum \bar s$, where the sum is over all $s\in \mathcal
T_\lambda$ obtained from $t\in \mathcal T_\lambda$ by exchanging any
$k$ entries of any column  with the top $k$ entries of the next
column, while maintaining the vertical order of each of the exchanged
sets.  There is a Garnir relation $g^t_{c,k}$ for every $t \in
\mathcal T_\lambda$, every column $c \in [\lambda_1-1]$,  and  every
$k$ from $1$ to the length of the column $c+1$.  Let $G^\lambda$ be
the subspace of $M^\lambda$ generated by these Garnir
relations. Clearly $G^\lambda$ is invariant under the action of
$S_m$. The  presentation of $S^\lambda$ obtained in Section 7.4 of \cite{Fu} is given by
   \be \label{Fultoneq} M^\lambda/G^\lambda \cong S^\lambda. \ee
   
 On page 102 (after Ex. 15) of \cite{Fu}, a presentation of $S^\lambda$ with fewer relations is given.   The presentation is
\be \label{1fulton} M^{\lambda} / G^{\lambda,1} \cong S^{\lambda} ,\ee where
 $G^{\lambda,1}$ is the subspace of $G^\lambda$ generated by $$\{ g^t_{c,1} : c \in [\lambda_1-1], t \in \mathcal T_\lambda \}  .$$ 
 
 In Appendix 1 of \cite{DI}, a proof that the   generalized Jacobi relations (\ref{type1}) are equivalent to the relations 
 \be\label{filipov} [[x_1, x_2, \ldots,  x_n], y_{1},\ldots , y_{n-1}]
=\sum_{i=1}^{n} [[x_1, x_{2}, \ldots , x_{i-1},y_{1}, x_{i+1}, \ldots , x_{n}],x_{i},y_{2},\ldots , y_{n-1}]
 \ee
% \begin{align}
%\label{filipov} &[[a_1, a_2, \ldots,  a_n], b_{1},\ldots , b_{n-1}]
%\\ \non&=\sum_{i=1}^{n} [[a_1, a_{2}, \ldots , a_{i-1},b_{1}, a_{i+1}, \ldots , a_{n}],a_{i},b_{2},\ldots , b_{n-1}]
%\end{align}
is given.\footnote{This equivalence can also be obtained as a consequence of Theorem~\ref{eigenth} and (\ref{1fulton}). 
 %In fact, in the original version of \cite{FHSW1}, we were unaware of (\ref{filipov}).
}
 This gives an alternative presentation of $\rho_{n,k}$ for all $n,k$. 

Using the natural correspondence between generators $[[a_1,\dots,a_n],b_1,\dots,b_{n-1}]$  of $V_{n,3}$ and column tabloids $\bar t$,  where $t$ is the tableau whose first column is $a_1,\dots,a_n$ and whose second column is $b_1,\dots, b_{n-1}$, we see that the alternative Jacobi relations (\ref{filipov}) correspond to the Garnir relation $g^t_{1,1}$ for $\lambda = 2^{n-1}1$.  Thus  the natural correspondence between generators yields an isomorphism from  $\rho_{n,3}$ to the realization of $S^{2^{n-1}1}$ given in (\ref{1fulton}). 
%generators gives an explicit isomorphism from  $\rho_{n,3}$ to $S^{2^{n-1}1}$.  

The natural  correspondence between generators of $V_{n,3}$ and generators of $M^{2^{n-1}1}$ also takes the generalized Jacobi relations (\ref{type1}) to the Garnir relations $g^t_{1,n-1}$.  This enables us to give another presentation of $S^{2^{n-1}1}$ with fewer relations than that of (\ref{Fultoneq}).  In fact, we can extend this to a wider class of Specht modules.
Indeed, the natural  correspondence between generators of $V_{n,3}$ and generators of $M^{2^{n-1}1}$ can  be used to transfer the operator $\varphi$ on $ V_{n,3}$ of 
Theorem~\ref{eigenth}  to $M^{2^{n-1}1}$.   For $d \in [n]$, a generalization $\varphi_{n,d}$ of $\varphi$ can be defined on 
$M^{2^d 1^{n-d}}$. We obtain a generalization of  Theorem~\ref{eigenth}  for $\varphi_{n,d}$ and use it to prove the following result,
which gives  a new presentation  with fewer relations than that of (\ref{Fultoneq}) for  Specht modules $S^\lambda$ whose conjugate shape $\lambda^\prime$ has strictly decreasing parts.\footnote{This result was obtained only for staircase shapes in an earlier version of \cite{FHSW1}.}

\begin{thm} \label{newpres}  For $\lambda \vdash m$, let $\mathcal T^*_\lambda$ be the set of  Young tableaux of shape $\lambda$ in which each element of $[m]$ appears once and the  columns increase,  and let $\tilde G^\lambda$ be the subspace  of $M^\lambda$  generated by $$\{ g^t_{c,\lambda^\prime_{c+1}} : c \in [\lambda_1-1], t \in \mathcal T^*_\lambda \}  .$$ 
If $\lambda^\prime$ has strictly decreasing parts then
$ S^\lambda  \cong M^\lambda / \tilde G^\lambda$.
\end{thm}  




In \cite{BF},  Brauner and Friedmann obtain a result analogous to the generalization of Theorem~\ref{eigenth} discussed above and use it to obtain an interesting new presentation of Specht modules of {\it all} shapes, in which the  number of relations has  been similarly reduced. The new presentation implies the presentation (\ref{1fulton}).  Another proof of Theorem~\ref{catalanke} is also discussed in \cite{BF}.

\printbibliography

\end{document}





\end{document}
